Response to reviewers’ comments

Reviewer 1: Dear Authors, the Manuscript "Diabetes Mellitus: An overview on the types, prevalence, comorbidity, complication, genetics, economic implication and treatment" is extensive but offers no novelty. Therefore, I don't find it applicable for the WJD.

Response: Contrary to your believe, we believe that the review manuscript will contribute to the available information on Diabetes Mellitus.

Reviewer 2: Many parts of the manuscript have no reference citation. Please check the whole manuscript accordingly. A recent systematic review and meta-analysis (10.1002/jmv.27996) documented that new-onset type 1 diabetes and ketoacidosis have increased worldwide during the pandemic period. This issue should be mentioned in ‘DIABETES IN CHILDREN AND ADOLESCENTS’ to explain the current review's importance. At least a summary graph is needed to explain the study's schematic.

Response: Many thanks for the comments. We have corrected the reference citations and added the systematic review as suggested in reference 9. We have added figure 1 to explain the study schematic.